Author's response to reviews

Title: Modeling emergency department visit patterns for infectious disease complaints: results and application to disease surveillance

Authors:

Judit C Brillman (jbrilman@salud.unm.edu)
Tom Burr (tburr@lanl.gov)
David Forslund (dwl@lanl.gov)
Edward Joyce (joyce@lanl.gov)
Rick Picard (picard@lanl.gov)
Edith T Umland (eumland@salud.unm.edu)

Version: 3 Date: 23 February 2005

Author's response to reviews:

To: Editors, BioMed Central.
From: Judith C. Brillman
Re: MS: 2693843934728047
Modeling Emergency Department Visit Patterns for Infectious Disease Complaints: Results and Application to Disease Surveillance
Date: February 22, 2005

Thank you for you acceptance in principle of the above manuscript. We have completed all the requested formatting changes.

Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).
Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. This has been done.
Country - Please remember to include the country at the end of all the affiliation details. USA added.
Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names. Each affiliation is given in full.
Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers.
For example:
Enitan Sawyerr1,2, Emma Veitch3
1 Plant Physiology, Zentrum fur Molekularbiologie der Pflanzen (ZMBP), Universitat Tubingen, Auf der Morgenstelle1, D-72076 Tubingen, Germany
2 Current address: Protein Interaction laboratory, Dept. Biochemistry and Molecular Biology, University of Southern Denmark, Campusvej 55, 5230 Odense, Denmark
3 Plant Biology, University of Minnesota, 220 Biological Sciences Center, 1445 Gortner Ave., St. Paul, MN55108-1095, USA
The numbering system has been adhered to.
Abstract - Please ensure this is identical, word-for-word, in the manuscript and in the submission system. Please change 'Conclusion' to 'Conclusions'. Done
Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). All caps except the initial cap have been removed.
Competing interests - If there are none to declare, please write 'The authors declare that they have no competing interests'. Wording was changed.
References - The reference list should contain all authors' names, regardless of how many there are. The term "et al" should not be used. All authors names are now included.
Tables - These should both be oriented in portrait rather than in landscape.

Tables are in portrait format.
Table 1 - Tables must be divided into cells / fields. The borders of tables included as part of the manuscript file must be visible black lines. Tables generated with tabbed text are not acceptable. Please ensure that text alignment in columns is clear and consistent.
Figure titles - Please place a full stop at the end of each figure title. Done.
Page 20 - Please remove this blank page. Done
Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Figures were cropped according to your guidelines.